Author’s response to reviews

Title: Effect of integrating maternal health services and family planning services on postpartum family planning behavior in Ethiopia: Results from a longitudinal survey

Authors:

Linnea Zimmerman (linnea.zimmerman@jhu.edu)
Yuanyuan Yi (yyi7@jhu.edu)
Mahari Yihdego (mahariyihdego@gmail.com)
Solomon Abrha (abrsole2006@gmail.com)
Solomon Shiferaw (solomon.shiferaw@gmail.com)
Assefa Seme (assefa.seme@gmail.com)
Saifuddin Ahmed (sahmed3@jhu.edu)

Version: 2 Date: 23 Sep 2019

Author’s response to reviews:

Please see our responses below. One additional note: I have attempted to link to my Orcid ID but because it is registered to a different email address I have been unable to. My Orcid ID is https://orcid.org/0000-0002-0118-0889

1. In order to adhere to our submission guidelines (https://bmcpublichealth.biomedcentral.com/submission-guidelines/preparing-your-manuscript/research-article), please include a Background section in the abstract. The background section was present but not labeled. It is now clearly differentiated in the abstract.

2. Please represent authors' names using their full initials, not their full name, in the Authors’ Contributions section. If there are any duplicated initials, please differentiate them to make it clear that the initials refer to separate authors. This has been addressed.

3. Appendix 1 has not been cited in the body of the manuscript. Please ensure that all tables, figures and additional files are cited in the body of the manuscript. We have removed Appendix 1 as it was originally intended to provide supplementary information but was removed from the final draft.
4. The tables in the manuscript have been duplicated, please ensure that only one copy of each table is attached. Please include only the tables included in the word document. The excel tables have been removed.

5. Please remove the STROBE checklist attachment, as it is no longer required at this stage. This has been removed.

6. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. If the manuscript text is duplicated in the manuscript file, please remove the duplicate text.